Author's response to reviews

Title: Distal left circumflex coronary artery flow reserve recorded by transthoracic Doppler echocardiography: a comparison with Doppler-wire

Authors:
Antonio Auriti (a.auriti@sanfilipponeri.roma.it)
Christian Pristipino (c.pristipino@sanfilipponeri.roma.it)
Cinzia Cianfrocca (c.cianfrocca@mclink.it)
Antonino Granatelli (a.granatelli@sanfilipponeri.roma.it)
Vincenzo Guido (vincenzo.guido@aliceposta.it)
Francesco Pelliccia (f.pelliccia@mclink.it)
Salvatore Greco (sa.gre@tiscali.it)
Giuseppe Richichi (g.richichi@sanfilipponeri.roma.it)
Massimo Santini (m.santini@rmnet.it)

Version: 4 Date: 9 June 2007

Author's response to reviews: see over
Modifications from previous version (2)

1) Authors’ contributions moved
2) AKNOWLEDGMENTS into Acknowledgments
3) Tab I into Table 1
4) title modified as on instructions
5) P4 L21 “…The operator was blinded to the results of TTDE”…. cancelled
6) P6 L1 ….(fig 1)…. added
7) Adenosine turned into adenosine when possible